Author's response to reviews

Title: Jejunoduodenal intussusception caused by a solitary polyp in a young female with Peutz-Jeger's syndrome: a case report

Authors:

Ali Ozer (draliozerr@gmail.com)
Pinar Sarkut (pinartasar@gmail.com)
Ersin Ozturk (drozturk@uludag.edu.tr)
Tuncay Yilmazlar (tunyil@uludag.edu.tr)

Version: 4 Date: 7 September 2013

Author's response to reviews:

We have revised our manuscript as you recommended. However, it was not possible to reach all data about patient. Because she lives in another city and also she was on vacation as we operated her. After her operation, we get limited data about her situation. Despite all, we made a substantive revision as possible. The following changes are made:

Case presentation:
We have added more details about history part.
The features of the physical examination, vital signs and laboratory are more elaborated.
Unfortunately we have not enough data for genetical history about her and her family.
We added the other CT findings as you wish.
Discussion:
This part is almost rewritten. We hope it has more educational value now.
References:
This part is corrected.
Figures:
We added labels on the pictures and we attached a histopathology slide as you wish.
Finally we have a professional service for editing.